

# G OPEN ACCESS

**Citation:** Miller L, Prieto Merino D, Baisley K, Hayes R (2022) Hidden heterogeneity: Uncovering patterns of adherence in microbicide trials for HIV prevention. PLoS ONE 17(5): e0267011. https://doi.org/10.1371/journal.pone.0267011

Editor: Patricia Evelyn Fast, IAVI, UNITED STATES

Received: June 2, 2021

Accepted: March 30, 2022

Published: May 12, 2022

**Copyright:** © 2022 Miller et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

Data Availability Statement: The data that support the findings of this study were kindly provided by the trial teams of the HIV Prevention Trials Network (HPTN) 035, the Microbicides Development Programme (MDP) 301, and the Carraguard trial at the Population Council. Restrictions apply to the availability of these data, which were provided under agreement for the current study, and so are not publicly available. Those interested in accessing the data from the clinical trials used in this study can contact the owners of the data directly to apply for access. Requests for HPTN 035 trial data can be sent to Statistical Center for HIV/AIDS Research and Prevention (SCHARP) at **RESEARCH ARTICLE** 

# Hidden heterogeneity: Uncovering patterns of adherence in microbicide trials for HIV prevention

## Lori Miller<sup>1\*</sup>, David Prieto Merino<sup>1,2</sup>, Kathy Baisley<sup>1</sup>, Richard Hayes<sup>1</sup>

1 London School of Hygiene and Tropical Medicine, London, United Kingdom, 2 Applied Statistical Methods in Medical Research Group, Catholic University of Murcia (UCAM), Murcia, Spain

\* Lori.Miller@alumni.lshtm.ac.uk

# Abstract

# Background

Interpretation of clinical trial results testing vaginal microbicide gels for HIV prevention depends on participant adherence. Prior to the era of antiretrovirals, microbicide trials collected adherence data via self-report, and trials typically reported trial population adherence as overall averages in primary results manuscripts. This study first sought to determine if different patterns of adherence from three trials of vaginal microbicide gels could be identified, using self-reported data and if so, how those patterns compare across trials. The second objective was to explore which individual-level factors were associated with different adherence patterns.

# Methods

Data from the following three clinical trials of vaginal microbicides were used for this study: HIV Prevention Trials Network (HPTN) 035 testing PRO 2000 and Buffergel, the Microbicides Development Programme (MDP) 301 testing PRO 2000, and the Population Council's Carraguard study, testing Carraguard gel. Latent Class Analysis (LCA) was used to identify longitudinal patterns of adherence using self-reported data about gel use. Multinomial multivariable logistic regression was used to estimate relative risk-ratios for factors which were independently associated with different latent adherence trajectories within each trial, and compared across trials.

# Results

Included in this analysis are 2,282 women from HPTN 035 (age 17–56 years), 6238 women from MDP 301 (age 16–75 years), and 6039 women from Carraguard (age 16–73 years). Using LCA, 3–4 different patterns of gel adherence were identified in each trial; these patterns were similar across the trials. Factors associated with adherence patterns were identified in all trials. Older age was associated with the adherence trajectory that consistently reported gel use in three trials. Participant-reported negative reaction of partners to the gel was associated with trajectories that reported less consistent adherence in two trials. A

sc.mtn-data-access@scharp.org. Requests for MDP 301 trial data can be sent to the Medical Research Council (MRC) Clinical Trials Unit at mrcctu.ctuenquiries@ucl.ac.uk. Requests for Carraguard trial data can be sent to Marlena Plagianos, the Population Council, at mplagianos@popcouncil.org.

Funding: This study was supported by the UK Economic and Social Research Council. Research reported in this publication was supported by the National Institute of Allergy And Infectious Diseases of the National Institutes of Health under Award Number T32AI114398. The content is solely the responsibility of the authors and does not necessarily represent the official views of the National Institutes of Health. RH and KB receive funding from the UK Medical Research Council (MRC) and the UK Department for International Development (DFID) under the MRC/DFID Concordat agreement, which is also part of the EDCTP2 programme supported by the European Union (Grant Ref: MR/R010161/1). The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

greater number of baseline-reported sex partners or sex acts was associated with trajectories which reported less consistent adherence in some trials. Trial site location was associated with membership of trajectories in all trials.

## Conclusion

LCA was able to identify patterns of microbicide gel adherence in clinical trials that used self-reported data. Key factors associated with patterns of adherence in this study were participant age, clinical trial site location, and partner reaction to the study gel. These findings, in particular, age and perceived partner reaction to the method, are consistent with results from other clinical trials and programmatic rollout of biomedical HIV prevention methods for women in Africa. This study contributes to the body of evidence that women need more support to navigate power dynamics within their relationships with men so that they can successfully use HIV prevention methods.

# Introduction

Understanding participant adherence to study product is critical to interpretation of results from clinical trials testing vaginal gel microbicides to prevent sexual transmission of HIV. As vaginal gel microbicides are user-controlled products, women may or may not use these topically-applied investigational products according to instructions. Therefore, a trial result that does not show a reduction in HIV incidence in the active gel arm might be due to an investigational product not having sufficient biological efficacy, or it might be due to low usage by trial participants. This is particularly a problem because trial participants often report good adherence whether or not they have used the gels consistently. The challenges of adherence–both of low adherence in trials [1–3] and, for trials relying on self-reported data [4–12], not knowing the cause of null results–have hampered the microbicide field over time [13]. While these challenges have motivated funders such as the National Institute of Allergy and Infectious Diseases to re-focus efforts on methods of HIV prevention such as vaginal rings and long-acting pre-exposure prophylaxis (PrEP) [14], women need a range of options [14–16]. Community stake-holders and researchers support continued development of on-demand products [14, 17].

While adherence is a key factor in understanding trial results, primary-results publications typically provide self-reported adherence to microbicides as overall averages for trial populations. These averages can create an impression of overall high adherence to the study product. In reality, each woman has her own trajectory of adherence during her follow-up, and a longitudinal approach to data analysis may provide a better understanding of how participants use study products over time. It would be helpful to understand if there are typical patterns of adherence that exist among trial participants, rather than adherence being characterised as static. The first objective of this study was to determine if different patterns of adherence using self-reported data from three trials of the effectiveness of coitally-dependent microbicide gels could be identified, and if so, how those patterns compare across trials. If patterns were identified, the second objective was to explore which individual-level factors were associated with different adherence patterns.

# Methods

Data from three completed non-antiretroviral vaginal-gel microbicide trials were included in this study: HIV Prevention Trials Network (HPTN) study of PRO2000 and BufferGel (HPTN035 [11]), Microbicides Development Programme's trial of PRO2000 (MDP301 [12]), and Population Council's trial of Carraguard (Carraguard [1]). Data for this type of analysis were not available from anti-retroviral (ARV) based microbicide clinical trials at the time this study commenced.

HPTN035 was a four-arm phase II/IIb randomised placebo-controlled trial testing 0.5% PRO 2000 gel and BufferGel for the prevention of sexual transmission of HIV against a placebo gel and a condom-only arm. A total of 3101 women were enrolled in eight sites in Malawi, South Africa, Zambia, Zimbabwe, and the US. Results indicated no preventive effect for BufferGel and a non-significant 30% reduction in HIV incidence for 0.5% PRO 2000.

MDP 301 was a three-arm phase III randomised, double-blind placebo-controlled trial testing 0.5% and 2% PRO2000 for the prevention of sexual transmission of HIV against a placebo gel. In total, 9385 women were enrolled in six sites in South Africa, Tanzania, Uganda, and Zambia. The 2% PRO 2000 arm was closed early due to futility. Trial results for 0.5% PRO2000 did not indicate an effect on HIV incidence.

Carraguard was a phase III randomised, placebo-controlled, double-blind trial to test the effect of Carraguard gel for the prevention of sexual transmission of HIV against a placebo gel. It enrolled 6202 women in three sites in South Africa. Carraguard was not shown to reduce HIV incidence in this trial.

All three trials tested vaginal-gels used just before sex. Self-reported "gel use at last sex act" was collected at monthly or quarterly visits over follow-up. <u>Table 1</u> provides information about each trial including investigational product, trial population size, age range of participants, locations, type and frequency of self-reported adherence data collected (Carraguard trial used a dye stain assay as its primary adherence measure), and trial dates.

Organization name Candidate product "Trial name"	Average adherence overall by self-report	Total number of participants	Age range of participants	Type of data	Frequency of adherence data collection	Trial dates	Locations	Planned follow-up	Actual follow- up and notes
HIV Prevention Trials Network (HPTN) BufferGel PRO 2000 "HPTN 035"	81%	3,101 (This analysis: 2,282)	17–56 years	Categorical: Gel use at last sex act (yes/ no)	Quarterly	February 2005 —September 2008	Malawi, South Africa, Zambia, Zimbabwe, USA	12–30 months	Average follow-up was 20.4 months; trial closed as planned
Microbicides Development Programme (MDP) PRO 2000 "MDP 301"	89%	9,385 (This analysis: 6,238) (includes 0.5% gel arm + placebo to 52 weeks)	16–75 years	Categorical: Gel use at last sex act (yes/ no)	Monthly	October 2005— September 2009 (2% arm dropped February 2008)	South Africa, Tanzania, Uganda, Zambia	12 months for primary analysis	For 0.5% PRO 2000 and placebo gel arms, follow- up was 12 months as planned
Population Council Carraguard "Carraguard"	96%	6,202 (This analysis: 6,038)	16–73 years	Categorical: Gel use at last sex act (yes/ no)	Quarterly	March 2004— March 2007	South Africa	9–24 months	Trial closed as planned

Table 1. Clinical trials included in the latent class analysis.

https://doi.org/10.1371/journal.pone.0267011.t001

### Identifying patterns of adherence: Latent class analysis

Latent class analysis [18] (LCA) was used to identify longitudinal patterns of adherence using self-reported data, referred to here as "latent adherence trajectories". Answers to the question "did you use the gel at your last sex act" asked monthly or quarterly, over time, were the basis for this analysis.

LCA is a type of finite mixture modelling, which is a family of modelling techniques that allow subgroups within a larger population to be identified. The goal of LCA is to take a seemingly homogeneous population and "un-mix" it so that existing constituent subgroups are identified. The latent subgroups are identified and defined by their similar response patterns to a particular set of questions or items. Because the answers to sets of questions are directly observable, it is their particular combination of response patterns that is then able to characterise the latent subgroup. Models with different numbers of trajectories are fitted, and an optimal model with a chosen number of trajectories is selected after assessing various criteria. There is no accepted best method for model selection, thus different investigators might come to different conclusions based on their evaluations of criteria and usefulness of the various models.

The selected LCA model will have a certain number of trajectories, and two parameters are estimated. The first parameter estimates the probability of a particular response to each item at each timepoint conditional on belonging to that particular latent trajectory (in the present study the estimate is for gel use at last sex act). The second parameter estimates the proportion of individuals in the population estimated to be in each latent trajectory.

Finally, the model provides estimates for each individual of their probability of belonging to each of the latent trajectories specified in the model, based on how the individual's observed response pattern compares to the item response probabilities estimated by the model; called the posterior probability.

For this analysis, the LCA Stata plugin [19] was used in Stata SE13 for LCA. All data-for individuals with both complete and incomplete data-were used in the analysis to model the latent trajectories, as the software uses a Full Information Maximum Likelihood [18], which assumes data are missing completely at random or missing at random.

In this analysis, models with 1–6 latent adherence trajectories were fitted for each of the three trials separately. To select the model considered to best describe adherence patterns for each trial, the following criteria were examined: trajectory shape, proportion of population belonging to each trajectory, trajectory separation, homogeneity, relative fit indices (Bayesian, adjusted Bayesian, Akaike information criteria), relative entropy, parsimony, interpretability, and model usefulness. As the objective of this analysis was to reduce a large volume of data to better understand adherence patterns, key aspects of model selection were parsimony and interpretability of the latent trajectory structures. More details about latent trajectory information for each trial and choice of model are provided in the S1 Appendix.

# Identifying factors associated with adherence patterns: Multinomial logistic regression

In this exploratory analysis examining if individual level factors might be associated with different adherence trajectories, multivariable multinomial logistic regression (Stata SE13) was used to estimate relative-risk-ratios (RRRs) and their 95% confidence intervals (CIs) for factors which were independently associated with different latent adherence trajectories within each trial. The latent adherence trajectories, derived from the LCA analysis, were based on participant answers to the question "did you use the gel at your last sex act" asked monthly or quarterly, over time. The associations between types of latent adherence trajectories and explanatory factors were then compared across trials. Latent adherence trajectory was the outcome variable, with regressions weighted by each trial participant's posterior probability of belonging to each of the latent trajectories, to account for uncertainty in class membership in the model.

Three types of individual-level factors were considered in this analysis: demographic characteristics (site, age, education), self-reported baseline behavioural characteristics (recent number of sex partners, recent number of sex acts, condom use, anal sex, transactional sex), and study exit acceptability characteristics (variables reflecting self-reported views of the microbicide gel and participants' reports of how they thought their partners viewed the gel). The RRRs represent the ratio of the probability of being in a particular adherence trajectory versus the reference trajectory, for one categorical group versus a reference categorical group.

For each trial, the age-adjusted association of each covariate of interest and the latent adherence trajectory was examined. Variables that showed evidence of an age-adjusted association with the latent adherence trajectory based on a likelihood ratio chi-square test of  $p \le 0.05$  were added to the multivariable multinomial logistic regression (the model) one at a time, starting with those that had the strongest evidence of an association. Each variable was added into the model and assessed using the likelihood ratio chi-square test statistic of the full model versus the model without the variable until a final set of variables were identified which had strong evidence of an association ( $p \le 0.05$ ) with the latent adherence trajectory membership.

#### **Ethical approval**

This study was approved by the London School of Hygiene and Tropical Medicine (LSHTM) Ethics Committee, reference 6491. Ethical approvals for included trials are reported elsewhere [1, 11, 12]. Datasets received were fully anonymised.

#### Results

#### **Participants**

In this analysis 2,282 women from HPTN 035 were included, ranging in age from 17–56 years. 1437 women had some secondary education or higher; 844 women had a primary education or below.

6238 women were included in this analysis from MDP 301, ranging in age from 16–75 years. 1508 women had some secondary education or higher; 4730 women had a primary education or below. 6039 women were included in this analysis from Carraguard, ranging in age from 16–73 years. 1142 women had some secondary education or higher; 455 women had a primary education or below.

#### Latent class model selection and patterns of adherence

A model with four adherence trajectories was chosen for HPTN035 and MDP301. A model with three adherence trajectories was chosen for Carraguard. For each trial, Fig 1 shows the pattern of each adherence trajectory, and the estimated proportion of trial participants in each trajectory.

For all three trials, the largest subgroup within the trial population was the subgroup that consistently had a high probability of reporting gel use at last sex act throughout trial followup ("Consistently-high" subgroup: 59% HPTN035, 70% MDP301, 91% Carraguard). In all three trials, this was also the subgroup that had the least variability in its gel use reporting. In all three trials, a subgroup was identified that initially had a high probability of reporting gel use, but about midway through the trial, probability of reporting gel use at last sex act decreased and remained low for the rest of the trial ("Later-decliners" subgroup: 20% HPTN035, 20% MDP301, 5% Carraguard). A third subgroup common to all trials had variable





Latent class membership prevalences n=6238					
	Consistently high	70%			
	Later decliners	9%			
	Variable	18%			
	Early decliners	3%			





https://doi.org/10.1371/journal.pone.0267011.g001

reporting of gel use with a somewhat up-and-down pattern, initially having a high probability of reporting gel use, then a lower probability of reporting gel use, followed by an increased probability of reporting gel use. ("Variable-adherers" subgroup: 17% HPTN035, 18% MDP301, 4% Carraguard). Finally, the subgroup "Early-decliners" was identified in two of the three trials (5% HPTN035, 3% MDP301), characterised by women who initially had a high probability of reporting gel use, but the probability of reporting gel use at last sex act decreased sharply after the first or second visit and remained lower than any of the other trajectories for the remainder of follow-up.

#### Factors associated with adherence patterns

The "Consistently-high" subgroup was used as the reference trajectory in the multinomial regression analysis for each trial (Tables 2–4).

**Age.** Across all trials (Tables 2–4) older age, versus younger age, was associated with belonging to the latent adherence trajectories which consistently reported gel use at last sex act ("consistently-high" group). In all trials, younger women had a greater chance than older women of belonging to an adherence trajectory that reported less consistent adherence to using the study gel at last sex act: they were more likely to be a part of the "Later-decliners", "Variable-adherers" or "Early-decliners" trajectories, compared to the consistently-high adherence-reporting group.

**Site.** In this study, model results consistently showed strong evidence across all trials that site was associated with latent adherence trajectory (Tables 2–4). The direction of the association depended on which sites were being compared.

**Reported negative reaction of partner to gel.** In the adjusted analysis, there was strong evidence that reported partner-dislike or refusal of gel was associated with an increased chance of being in an adherence trajectory that did not consistently report gel use in two of the three trials included in this study (HPTN035, Table 2; Carraguard, Table 4). For example, in HPTN035, women who reported that their partner disliked the gel, compared to those who didn't, were over 5 times more likely to be in the early-decliners group than in the consistently-high group (aRRR = 5.53, CI = 2.23–13.75), and around 3 times more likely to be in the later-decliners or variable-adherence groups, compared with women who did not report partner-dislike of gel. Similarly, in Carraguard, women who reported that their partner refused them to use gel were around 3 times more likely to be in the later-declining group compared to the consistently-high adhering group, and around 4 times more likely to be in the variable-adherence group, compared to women who provided other responses. In MDP301, in the age-adjusted analysis, reported partner-dislike of gel was associated with an increased chance of belonging to an adherence trajectory that did not consistently report gel use; however this association was not evident in the fully-adjusted analysis.

Number of sex partners reported at baseline. In the Carraguard trial, there was strong evidence that reported higher number of recent sex partners at baseline was associated with belonging to latent trajectories that did not consistently report gel use (Table 4). Women who had 2 or more partners were nearly 2 times more likely to be in the later-decliner or variable-adherence groups than in the consistently-high group (aRRR = 1.56, CI = 1.06-2.29 and aRRR = 1.74, CI = 1.13-3.67), compared to women who reported only 1 partner at baseline. In the other trials, although women with a higher number of partners at baseline were more likely to belong to an adherence group that did not consistently report gel use, the association was not statistically significant.

**Number of sex acts reported at baseline.** In the MDP301 trial, women who reported a higher number of recent sex acts at baseline had a higher chance of belonging to an adherence

#### Table 2. HPTN 035 multivariable multinomial logistic regression.

HPTN 035 multivariable multinomial logistic regression					
	Reference Class: Reported consistently high adherence				
	Early decliners	Variable	Later decliners		
n = 2282	Adjusted RRR	Adjusted RRR	Adjusted RRR	Likelihood ratio chi square test	
	[95% CI]	[95% CI]	[95% CI]		
Age	2.31	1.62	1.64	p<0.001	
Under 30 vs. 30+	[1.34-4.00]	[1.21-2.17]	[1.25-2.15]		
Site (vs. Durban)				p<0.001	
Philadelphia	2.15	0.96	1.01		
	[0.98-4.71]	[0.55-1.68]	[0.61–1.67]		
Harare	0.39	0.40	0.74		
	[0.13-1.20]	[0.22-0.74]	[0.47-1.16]		
Chitungwiza	0.24	0.24	0.55		
	[0.07-0.82]	[0.12-0.47]	[0.35-0.87]		
Hlabisa	0.83	0.94	0.72		
	[0.39-1.73]	[0.62-1.42]	[0.48-1.09]		
Blantyre	0.62	0.85	0.65		
	[0.30-1.29]	[0.58-1.26]	[0.44-0.95]		
Lilongwe	1.46	1.51	1.22		
	[0.80-2.64]	[1.06-2.14]	[0.87-1.70]		
Lusaka	0.66	0.63	0.80		
	[0.30-1.44]	[0.39-1.00]	[0.54-1.19]		
Partner dislike of gel	5.53	3.37	2.57	p<0.001	
He did not like it vs. other responses	[2.23-13.75]	[1.67-6.79]	[1.25-5.32]		
Education	0.87	1.17	0.95	p = 0.711	
Primary or less vs. secondary+	[0.47-1.59]	[0.83-1.64]	[0.69–1.30]		
Number of sex partners at baseline	1.68	1.51	1.47	p = 0.503	
2–6 vs. 1 in last 3 months	[0.60-4.71]	[0.72-3.13]	[0.75-2.87]		
Number of vaginal sex acts at baseline	1.43	1.29	0.96	p = 0.308	
5–21 vs. 0–4 in past week	[0.81-2.54]	[0.92-1.82]	[0.70-1.33]		
Condom use at baseline	0.86	0.88	0.81	p = 0.361	
No vs. yes to condom at last sex act	[0.55-1.34]	[0.68-1.14]	[0.64-1.04]		
Anal sex at baseline	2.52	1.36	1.82	p = 0.103	
Yes to anal sex ever vs. no	[1.02-6.21]	[0.67-2.73]	[0.99-3.34]		
Exchanged sex for money, etc. at baseline	0.31	0.44	0.73	p = 0.305	
Yes vs. no	[0.03-3.04]	[0.14-1.38]	[0.32-1.66]		
Participant thought gel was messy	1.07	1.17	1.19	p = 0.857	
Yes vs. other responses	[0.50-2.32]	[0.72-1.89]	[0.76-1.85]		
Participant thought the gel was difficult to remember	1.26	0.90	0.83	p = 0.694	
Yes vs. other responses	[0.64-2.49]	[0.57-1.42]	[0.53-1.30]		
Participant liked the gel because they thought it may protect against HIV	1.32	0.96	1.05	p = 0.672	
No vs. yes	[0.82-2.14]	[0.71-1.30]	[0.80-1.37]		
Participant reported that their partner liked the gel the last time they used it	1.10	1.14	1.18	p = 0.555	
Other responses vs. yes he liked it	[0.68-1.79]	[0.87-1.50]	[0.92-1.51]		

Variables at the top of the table show an association with latent adherence trajectory at the p = 0.05 level based on the likelihood ratio chi-square-ratio test and are adjusted for each other. Variables at the bottom of the table did not meet the criterion for being included in the model as having been associated with the latent adherence trajectories, but are included so that they may be compared across the trials. Variables in this section are adjusted for the variables which were associated with latent adherence trajectory, but not adjusted for the other variables which did not meet the criterion of p = 0.05 for the likelihood ratio chi-square test.

https://doi.org/10.1371/journal.pone.0267011.t002

#### Table 3. MDP 301 multivariable multinomial logistic regression.

MDP 301 multivariable multinomial logistic regression					
	Reference Class: Reported consistently high adherence				
	Early decliners	Variable	Later decliners		
n = 5083	Adjusted RRR	Adjusted RRR	Adjusted RRR	Likelihood ratio chi square test	
	[95% CI]	[95% CI]	[95% CI]		
Age	1.80	1.21	1.53	p<0.001	
Under 30 vs.30+	[1.25-2.60]	[1.04-1.42]	[1.24-1.88]		
Site vs. Durban				p<0.001	
Joburg	2.24	1.77	1.25		
	[1.50-3.33]	[1.47-2.14]	[0.98-1.60]		
Masaka	0.49	1.05	0.62		
	[0.19–1.28]	[0.77-1.43]	[0.39-0.99]		
Mwanza	1.00	0.92	0.96		
	[0.52-1.91]	[0.69-1.23]	[0.67-1.37]		
Africa Centre	0.77	0.72	0.79		
	[0.38-1.57]	[0.53-0.97]	[0.54-1.15]		
Mazabuka	0.46	0.88	0.91		
	[0.22-0.95]	[0.68-1.13]	[0.66-1.25]		
Number of sex acts at baseline	1.28	1.27	1.16	p = 0.017	
4+ or more last week vs. 3 or less	[0.91-1.81]	[1.08-1.48]	[0.94-1.44]		
Education	0.79	0.76	0.79	p = 0.009	
Primary or less vs. secondary +	[0.55-1.14]	[0.64-0.91]	[0.63-1.00]		
Number of sex partners at baseline	1.67	1.19	1.62	p = 0.757	
>1 partner in last week vs. 1	[0.35-7.98]	[0.51-2.78]	0.62-4.25]		
Condom use at baseline	0.92	0.88	1.05	p = 0.329	
Yes vs. no condom at last sex	[0.66-1.30]	[0.75-1.02]	[0.85-1.29]		
Anal sex at baseline	2.21	1.33	0.81	p = 0.397	
Yes anal sex in last 4 weeks vs. no	[0.81-6.04]	[0.72-2.45]	[0.30-2.17]		
Recommend gel to friends if halves women's risk of HIV?	0.90	0.86	1.14	p = 0.900	
Discourage vs. encourage	[0.27-3.01]	[0.51-1.46]	[0.62-2.10]		
Was it easy to predict when you needed to use the gel?	0.57	0.89	0.84	p = 0.167	
No vs. yes	[0.34-0.96]	[0.69–1.16]	[0.59–1.18]		
How likely do you think it is that you might get infected with HIV?				p = 0.620	
Not very likely vs. very likely	0.63	1.00	0.94		
	[0.39-1.02]	[0.80-1.25]	[0.71-1.27]		
Impossible vs. very likely	0.63	0.95	1.04		
	[0.33-1.19]	[0.72-1.24]	[0.72-1.48]		
Effect of gel on partner's enjoyment of sex?	1.08	1.00	0.92	p = 0.990	
Less enjoyable vs. other responses	[0.42-2.78]	[0.67-1.48]	[0.53-1.60]		
Partner dislike of gel	2.01	1.50	1.26	p = 0.165	
Disliked it vs. other responses	[0.87-4.64]	[0.98-2.29]	[0.68-2.33]		

Variables at the top of the table show an association with latent adherence trajectory at the p = 0.05 level based on the likelihood ratio chi-square-ratio test and are adjusted for each other. Variables at the bottom of the table did not meet the criterion for being included in the model as having been associated with the latent adherence trajectories, but are included so that they may be compared across the trials. Variables in this section are adjusted for the variables which were associated with latent adherence trajectory, but not adjusted for the other variables which did not meet the criterion of p = 0.05 for the likelihood ratio chi-square test.

https://doi.org/10.1371/journal.pone.0267011.t003

#### Table 4. Carraguard multivariable multinomial logistic regression.

Carraguard multivariable multinomial logistic regression					
	Reference Class: Re high ad	ported consistently herence			
	Later decliners	Variable			
n = 6039	Adjusted RRR	Adjusted RRR	Likelihood ratio chi square test		
	[95% CI]	[95% CI]			
Age	1.63	1.40	p < .001		
Under 35 vs. 35+	[1.26-2.11]	[1.06-1.86]			
Site			p < .001		
Medunsa (vs. UCT)	0.48	0.39			
	[0.34-0.66]	[0.27-0.55]			
MRC (vs. UCT)	1.94	1.24			
	[1.49-2.52]	[0.92-1.66]			
Number of sex partners at baseline	1.56	1.74	p = 0.008		
2 or more vs. 1	[1.06-2.29]	[1.13-2.67]			
What was your most recent partner's reaction to you using the study gel?*	3.26	3.98	p = 0.016		
Refused me to use gel vs. other responses	[1.22-8.69]	[1.30-12.17]			
Education*	0.79	0.82	p = 0.621		
Primary or less vs. secondary +	[0.44 - 1.40]	[0.42-1.60]			
Number of sex acts baseline	1.08	1.11	p = 0.677		
5 or more sex acts in past 2 weeks vs. 4 or less	[0.83-1.41]	[0.83-1.48]			
Condom use at baseline	0.99	0.91	p = 0.882		
No condom used at last sex act with steady partner vs. yes condom	[0.71-1.39]	[0.62-1.33]			
Anal sex at baseline	0.94	1.24	p = 0.844		
Yes unprotected anal sex in the past 3 months vs. no	[0.47-1.89]	[0.58-2.62]			
Have you ever had sex in exchange for money?	0.99	0.72	p = 0.773		
yes vs no	[0.48-2.06]	[0.27-1.86]			
What effect did the study gel have on your sexual pleasure?*			p = 0.531		
Less pleasure vs. more pleasure	2.20	0.44			
	[0.77-6.29]	[0.04-4.91]			
No effect vs. more pleasure	1.09	1.18			
	0.67-1.79	[0.67-2.09]			
What effect did the gel have on your most recent partner's sexual pleasure?*	1.38	0.99	p = 0.859		
Less pleasure vs. other responses	[0.46-4.10]	[0.24-4.09]			

\*Subsample of 1191 participants.

Variables at the top of the table show an association with latent adherence trajectory at the p = 0.05 level based on the likelihood ratio chi-square-ratio test and are adjusted for each other. Variables at the bottom of the table did not meet the criterion for being included in the model as having been associated with the latent adherence trajectories, but are included so that they may be compared across the trials. Variables in this section are adjusted for the variables which were associated with latent adherence trajectory, but not adjusted for the other variables which did not meet the criterion of p = 0.05 for the likelihood ratio chi-square test.

https://doi.org/10.1371/journal.pone.0267011.t004

trajectory that did not consistently report gel use (Table 3), compared to women who reported fewer sex acts. Women who had  $\geq$ 4 sex acts were about 1.3 times more likely to be in the early-decliner or variable-adherence groups than in the consistently high group, and about 1.2 times more likely to be in the late-decliner group, compared with women who reported 3 or less sex acts. In other trials, there was a trend for women who reported a higher number of sex acts to have an increased chance of being in an adherence trajectory which did not consistently report using gel: however, effect sizes were generally small, and the association was not statistically significant (Tables 2 & 4). **Education.** In MDP301, there was strong evidence that women with less education had an increased chance of membership of the adherence trajectory that consistently reported gel use at last sex act. (Table 3).

Women with only primary education or less were around 20% less likely to be in the earlydecliner, variable, or late-decliner adherence groups, than in the consistently-high adherence group, compared with women with secondary education or higher.

A similar trend of less education being associated with an increased chance of belonging to the adherence group which reported consistent gel use was seen in the Caraguard trial and less so in HPTN035 (Tables 2 & 4).

# Discussion

## Patterns of adherence

Latent class analysis identified subgroups of participants with different adherence patterns in all three trials in this exploratory study, using self-reported data. Latent adherence trajectories across the trials were similar. Common to all three trials were larger subgroups of participants who reported "Consistently-high" adherence, a subgroup of "Later-decliners" and a subgroup of "Variable-adherers." A small subgroup characterised as "Early-decliners" was identified in two of the three trials.

#### Factors associated with adherence patterns

There was strong evidence that site was associated with latent adherence trajectory membership in all three of the included trials. This association has been observed in other studies from the included trials [20–22] and other HIV prevention trials such as MIRA [23], and FEM-PrEP [24]. This finding indicates that local culture and site staff factors may have played a role in how participants reported their adherence or may have influenced participants' actual adherence. In some locations, underlying culture may influence participants to report higher adherence to "please" site staff, or participants may report gel use in order to avoid being reprimanded, removed from the trial, or showing vulnerability [25–30].

Younger age, in all three trials, was found to be associated with the subgroups of participants that did not consistently report high adherence; this finding has been observed in other HIV prevention studies [3, 21–23, 31–33]. Reasons that might affect younger women's ability to consistently use the gel could be that they are less likely to be in stable relationships, that they have more partners, and that they may be less able to plan sex. Older women may be more aware of their HIV risk, recognising non-faithful partners. Older women may be more likely to be mothers, and more interested in protecting themselves and their children from HIV. They may be living in more stable households where logistically they can better manage gel use, compared to younger participants who may be having sex outside of locations where they reside [3]. Older participants, compared to younger participants, may also have more selfefficacy and be better prepared to negotiate gel use with their partners or decide to use it even without partner approval.

It is not surprising that if a participant reports her partner did not like the gel or refused her use of the gel, her adherence to gel would be negatively affected. While these data were collected at study exit, after participants reported their adherence, these results are consistent with the growing body of evidence showing partner dynamics are a critical factor affecting female participants' adherence in biomedical HIV prevention trials as well as affecting wom-en's concerns about PrEP use [22, 25, 28, 29, 34–43].

The impetus for vaginal microbicide development came from the reality that many male partners refuse to use condoms and women need an HIV prevention strategy they can control, and yet this same dynamic continues to be a barrier for female participants to use the very product being tested to free themselves from that constraint. This speaks to the real and ongoing challenges that women face in sexual relationships with men. HIV prevention trial teams, with the consent of female participants, can design more comprehensive and systematic opportunities to engage male partners at the beginning of trials and throughout follow-up [22, 34, 39]. Clinical trials and programmes rolling-out proven HIV prevention methods will benefit from greater understanding of these dynamics, providing more support to women to effectively communicate with their male partners, ultimately empowering women to use HIV prevention products more effectively.

Women who reported a higher number of sex acts at baseline in MDP301 and a higher number of sex partners at baseline in Carraguard were associated with membership of the latent trajectories that did not consistently report high adherence. These results are in agreement with trial teams' own findings in other analyses that women at greater risk of HIV have less consistent adherence [20, 21]. Those who report more sex and sex partners at baseline might be exchanging sex, living or working in less stable environments, and be more mobile, all of which may make it challenging to use gel at each sex act. Participants with fewer sex partners and sex acts at baseline might represent a population of women in more stable relationships, with more established home situations, including locations for gel storage.

There was strong evidence that less education was associated with membership of the trajectory that consistently reported high adherence in MDP301. This was in agreement with MDP301's own findings when looking at predictors of consistent adherence [21]. Results from the other trials, however, were variable. There is not clear evidence on how education may be related to reported adherence. Possibly women with less education have less economic stability, may fear removal from the trial and loss of reimbursements, and thus might be more inclined to report "good" adherence. Also possible is that participants with more education feel freer to not follow directions from clinic staff or freer to report their actual adherence compared to women with less education.

# Latent class analysis as a tool for understanding adherence to HIV prevention products

In this study, the use of LCA was effective at identifying different patterns of adherence among a seemingly homogenous population of microbicide gel trial participants. This information was obtained with self-reported data, which is less expensive than biomarker data, and preserves clinical trial blinding. Identifying adherence trajectories and associated factors can help trial teams better understand their participant populations and better address suboptimal adherence issues.

For example, early-decliners may represent a proportion of women who joined the trial without the intention of using the study gel. Alternatively, they may have found using the gel initially difficult and may have benefited from targeted support on product use. Later-decliners may represent participants who stop using the gel for several reasons–such as real-life barriers that make gel use around the time of sex difficult and may represent women who would have benefitted from more intensive support around how to communicate with their male partners about the gel. Low reported adherence at one site might indicate significant barriers to gel use, and the need for understanding and engagement around cultural barriers to adherence. Very high reported adherence at one site might indicate better adherence, or could indicate that staff have poor rapport with participants, such that participants are not comfortable being honest. Trial teams can use such information to investigate circumstances, liaise with stakeholders to understand their needs, and re-train staff where necessary. Information learned through

follow-up can be integrated into trial procedures prospectively during implementation or for future studies to better support participant recruitment, product adherence, adherence reporting, and trial conduct, as well as planning and rollout of programmes for approved biomedical HIV prevention methods.

## Limitations

There are several limitations of this study. All adherence and behavioural data are selfreported. Data may be biased towards high adherence as participants may have been motivated to please staff members or feared removal from the trial [25–27, 29, 30, 40, 44]. The Carraguard trial asked only one question about gel use (the primary adherence measure was not self-report), which may have biased results towards higher self-reported adherence. Certain sexual behaviours may be underreported for fear of judgement by staff [45, 46]. Participants reported their perceived partner views of gels at study exit, after reporting their own adherence. Theoretically, women could have decided to state that their partners had a negative view of the gels as an "excuse" for their own non-adherence. If that occurred to a large extent, that could produce an association between low adherence and partner dislike of the gels that was stronger than in reality.

In this analysis, factors such as number of sex partners, condom use, and transactional sex at baseline were examined in relation to latent adherence trajectories. In reality, factors such as these may have changed over the course of trial participation. The multivariable models were built using a p-value threshold of 0.05 for the likelihood ratio chi-square test. This method has a high risk for type II error, particularly if a candidate factor is confounded in an equal and opposite direction by another factor not in the model.

The data included in this analysis were collected from trials that ended most recently in 2009. Thus, these data are from a period of time prior to antiretroviral microbicides. Since these trials were conducted, the field of biomedical HIV prevention has moved on to antiretrovirals and products that can be used for longer periods of time, such as vaginal rings and injectable pre-exposure prophylaxis (PrEP).

In this analysis, LCA was used to better understand longitudinal reporting patterns of gel use among women in the included microbicide trials. In LCA, there is no accepted best method for selecting the number of latent classes in a particular analysis, therefore different investigators might come to different conclusions based on their evaluations of criteria and usefulness of the various models. Once a model has been selected with a specified number of latent classes, the model will mathematically separate the population into the specified number of trajectories. LCA is a model which simplifies large amounts of data. In reality, each participant has her own trajectory of gel use over time. While the analysis is a simplification of reality, this simplification has supported interpretation of adherence in these trial populations.

# Conclusion

LCA combined with multinomial logistic regression is a method HIV prevention trials and programmes rolling out proven HIV prevention methods can use to interpret adherence data from their populations. This information can support identification of issues related to product use. Trial teams and programme implementers can then engage with communities to better understand barriers to adherence and develop strategies together to support optimal use of these methods [47]. An important factor associated with lower adherence in two trials in this study was perceived negative reaction of male partners to the study gel. These findings are consistent with results from other clinical trials and programmatic rollout of biomedical HIV prevention methods which highlight concerns expressed by women in Africa about the impact of

relationship dynamics when considering or using HIV prevention methods. This study contributes to the body of evidence that women need more support to navigate power dynamics within their relationships with male partners so that they can successfully use HIV prevention methods.

# Supporting information

**S1** Appendix. Latent adherence trajectory information and model selection for 3 trials. (DOCX)

# Acknowledgments

LM conceived and designed the study, conducted the analysis, and wrote the manuscript. RH, DPM, and KB provided statistical advice and critical review of the study and manuscript. All authors approved the final manuscript. The authors thank the trial teams for access to the data used for this study: HPTN035, MDP301, and Carraguard of the Population Council. The authors also thank the women who participated in these trials.

# **Author Contributions**

Conceptualization: Lori Miller.

Data curation: Lori Miller.

Formal analysis: Lori Miller.

Funding acquisition: Lori Miller.

Investigation: Lori Miller.

Methodology: Lori Miller, David Prieto Merino, Kathy Baisley, Richard Hayes.

Project administration: Lori Miller.

Supervision: Lori Miller, David Prieto Merino, Kathy Baisley, Richard Hayes.

Writing – original draft: Lori Miller.

Writing - review & editing: Lori Miller, David Prieto Merino, Kathy Baisley, Richard Hayes.

#### References

- Skoler-Karpoff S, Ramjee G, Ahmed K, Altini L, Plagianos MG, Friedland B, et al. Efficacy of Carraguard for prevention of HIV infection in women in South Africa: a randomised, double-blind, placebo-controlled trial. Lancet. 2008 Dec 6; 372(9654):1977–87. https://doi.org/10.1016/S0140-6736(08)61842-5 PMID: 19059048
- Marrazzo J, Ramjee G, Richardson BA, Gomez K, et al. Tenofovir-Based Preexposure Prophylaxis for HIV Infection among African Women. N Engl J Med. 2015; 372(6):509–18. https://doi.org/10.1056/ NEJMoa1402269 PMID: 25651245
- 3. Rees H, Delany-moretlwe S, Lombard C, Baron D, Panchia R, Myer L, et al. FACTS 001 Phase III Trial of Pericoital Tenofovir 1% Gel for HIV Prevention in Women. In: Conference on Retroviruses and Opportunistic Infections. Seattle, USA; 2015. p. 26LB.
- Kreiss J, Ngugi E, Holmes K, Ndinya-Achola J, Waiyaki P, Roberts PL, et al. Efficacy of nonoxynol 9 contraceptive sponge use in preventing heterosexual acquisition of HIV in Nairobi prostitutes. JAMA. 1989; 268(4):477–82.
- Roddy R, Zekeng L, Ryan K, Tamoufe U, Weir S, Wong E. A controlled trial of nonoxynol 9 film to reduce male-to female transmission of sexually transmitted diseases. N Engl J Med. 1998; 339:504–10. https://doi.org/10.1056/NEJM199808203390803 PMID: 9709043

- Richardson B, Lavreys L, Martin H, Stevens C, Ngugi E, Mandaliya K, et al. Evaluation of a Low-Dose Nonoxynol-9 Gel for the Prevention of Sexually Transmitted Diseases. Sex Transm Dis. 2001; 28 (7):394–400. https://doi.org/10.1097/00007435-200107000-00006 PMID: 11460023
- Van Damme L, Ramjee G, Alary M, Vuylsteke B, Chandeying V, Rees H, et al. Effectiveness of COL-1492, a nonoxynol-9 vaginal gel, on HIV-1 transmission in female sex workers: a randomised controlled trial. Lancet. 2002 Sep 28; 360(9338):971–7. <u>https://doi.org/10.1016/s0140-6736(02)11079-8</u> PMID: 12383665
- Peterson L, Nanda K, Opoku BK, Ampofo WK, Owusu-Amoako M, Boakye AY, et al. SAVVY (C31G) gel for prevention of HIV infection in women: a Phase 3, double-blind, randomized, placebo-controlled trial in Ghana. PLoS One. 2007 Jan; 2(12):e1312. https://doi.org/10.1371/journal.pone.0001312 PMID: 18091987
- Feldblum PJ, Adeiga A, Bakare R, Wevill S, Lendvay A, Obadaki F, et al. SAVVY vaginal gel (C31G) for prevention of HIV infection: a randomized controlled trial in Nigeria. PLoS One. 2008 Jan; 3(1):e1474. https://doi.org/10.1371/journal.pone.0001474 PMID: 18213382
- Halpern V, Ogunsola F, Obunge O, Wang C-H, Onyejepu N, Oduyebo O, et al. Effectiveness of cellulose sulfate vaginal gel for the prevention of HIV infection: results of a Phase III trial in Nigeria. PLoS One. 2008 Jan; 3(11):e3784. https://doi.org/10.1371/journal.pone.0003784 PMID: 19023429
- Abdool Karim SS, Richardson B a, Ramjee G, Hoffman IF, Chirenje ZM, Taha T, et al. Safety and effectiveness of BufferGel and 0.5% PRO2000 gel for the prevention of HIV infection in women. AIDS. 2011 Apr 24; 25(7):957–66. https://doi.org/10.1097/QAD.0b013e32834541d9 PMID: 21330907
- McCormack S, Ramjee G, Kamali A, Rees H, Crook AM, Gafos M, et al. PRO2000 vaginal gel for prevention of HIV-1 infection (Microbicides Development Programme 301): a phase 3, randomised, double-blind, parallel-group trial. Lancet. 2010 Oct 16; 376(9749):1329–37. <u>https://doi.org/10.1016/S0140-6736(10)61086-0 PMID: 20851460</u>
- Institute of Medicine. Methodological challenges in biomedical HIV prevention trials, IOM (Institute of Medicine). Washington DC: National Academies Press; 2008.
- 14. Pickett J, LeBlanc M-A. Whose choice is it anyway? International rectal microbicide advocates (IRMA). Chicago; 2018.
- Montgomery E, Beksinska M, Mgodi N, Schwartz J, Weinrib R, Browne E, et al. Preference and Choice of Four Vaginal HIV Prevention Placebo Dosage Forms Among Young African Women: Results of the Quatro Randomized Crossover Trial. In Madrid: HIV Research for Prevention conference (HIVR4P 2018); p. abstract OA05.04.
- Quaife M, Eakle R, Cabrera Escobar MA, Vickerman P, Kilbourne-Brook M, Mvundura M, et al. Divergent Preferences for HIV Prevention: A Discrete Choice Experiment for Multipurpose HIV Prevention Products in South Africa. Med Decis Mak. 2018; 38(1):120–33. https://doi.org/10.1177/ 0272989X17729376 PMID: 28863752
- Baeten JM, Hendrix CW, Hillier SL. Topical Microbicides in HIV Prevention: State of the Promise. Annu Rev Med. 2020; 71(1):361–77. https://doi.org/10.1146/annurev-med-090518-093731 PMID: 31613684
- Collins LM, Lanza ST. Latent Class and Latent Transition Analysis. Hoboken, New Jersey: John Wiley and Sons Inc.; 2010.
- Lanza S, Dziak JJ, Huang L, Wagner A, Collins L. LCA Stata Plugin Users' Guide Version 1.1. University Park; 2014.
- 20. Friedland BA, Stoner M, Chau MM, Plagianos MG, Govender S, Morar N, et al. Baseline Predictors of High Adherence to a Coitally Dependent Microbicide Gel Based on an Objective Marker of Use: Findings from the Carraguard Phase 3 Trial. AIDS Behav. 2015; 20:2565–2577
- Crook AM, Abaasa A, Chisembele M, Gafos M, Harvey S, Kaganson N, et al. Investigating predictors of consistent gel use in the MDP 301 microbicide trial. In: XVIII International AIDS Conference, Vienna. 2010. p. MO PE0389.
- Gorbach P, Kelly C, Borgerding J, Ramjee G, et al. Effects of Partnership Change on Microbicide Gel Adherence in HPTN 035. AIDS Behav. 2014; 18(5):855–61. https://doi.org/10.1007/s10461-013-0651-5 PMID: 24158488
- van der Straten A, Shiboski S, Montgomery ET, Moore J, De Bruyn G, Ramjee G, et al. Patterns and predictors of adherence to diaphragm use in a phase III trial in sub-Saharan Africa: a trajectory analysis. J Acquir Immune Defic Syndr. 2009 Apr 1; 50(4):419–26. https://doi.org/10.1097/QAI. 0b013e3181958511 PMID: 19214115
- 24. Van Damme L, Corneli A, Ahmed K, Agot K, Lombaard J, Kapiga S, et al. Preexposure prophylaxis for HIV infection among African women. N Engl J Med. 2012; 367(5):411–22. <u>https://doi.org/10.1056/</u> NEJMoa1202614 PMID: 22784040

- Corneli AL, McKenna K, Perry B, Ahmed K, Agot K, Malamatsho F, et al. The science of being a study participant: FEM-PrEP participants' explanations for overreporting adherence to the study pills and for the whereabouts of unused pills. J Acquir Immune Defic Syndr. 2015; 68(5):578–84. https://doi.org/10. 1097/QAI.0000000000525 PMID: 25761233
- van der Straten A, Montgomery ET, Musara P, Etima J, Naidoo S, Laborde N, et al. Disclosure of pharmacokinetic drug results to understand nonadherence. AIDS. 2015; 29(16):2161–71. <a href="https://doi.org/10.1097/QAD.0000000000801">https://doi.org/10.1097/QAD.0000000000801</a> PMID: 26544581
- Magazi B, Stadler J, Delany-Moretlwe S, Montgomery E, Mathebula F, Hartmann M, et al. Influences on visit retention in clinical trials: insights from qualitative research during the VOICE trial in Johannesburg, South Africa. BMC Womens Health. 2014 Jan; 14(1):88. <u>https://doi.org/10.1186/1472-6874-14-88 PMID: 25065834</u>
- van der Straten A, Stadler J, Montgomery E, Hartmann M, Magazi B, Mathebula F, et al. Women's experiences with oral and vaginal pre-exposure prophylaxis: the VOICE-C qualitative study in Johannesburg, South Africa. PLoS One. 2014 Jan; 9(2):e89118. <u>https://doi.org/10.1371/journal.pone.</u> 0089118 PMID: 24586534
- Corneli A, Perry B, Mckenna K, Agot K, Ahmed K, Taylor J, et al. Participants' explanations for nonadherence in the FEM-PrEP clinical trial. J of Acquired Immune Deficiency Syndromes. 2016; 71(4): 452– 61. https://doi.org/10.1097/QAI.0000000000880 PMID: 26536315
- Saethre E, Stadler J. Malicious whites, greedy women, and virtuous volunteers: negotiating social relations through clinical trial narratives in South Africa. Med Anthropol Q. 2013 Mar; 27(1):103–20. https:// doi.org/10.1111/mag.12018 PMID: 23674325
- Marrazzo J, Ramjee G, Nair G, Palanee T, Mkhize B, Nakabiito C, et al. Pre-exposure Prophylaxis for HIV in Women: Daily Oral Tenofovir, Oral Tenofovir/Emtricitabine, or Vaginal Tenofovir Gel in the VOICE Study (MTN 003). In: Conference on Retroviruses and Opportunistic Infections. 2013. p. Abstract# 26LB.
- Baeten JM, Palanee-Phillips T, Brown ER, Schwartz K, Soto-Torres LE, Govender V, et al. Use of a Vaginal Ring Containing Dapivirine for HIV-1 Prevention in Women. N Engl J Med. 2016; 375:2121– 2132 https://doi.org/10.1056/NEJMoa1506110 PMID: 26900902
- **33.** Nel A, Kapiga S, Bekker L-G, Devlin B, Borremans M, Rosenberg Z. Safety and Efficacy of Dapivirine Vaginal Ring for HIV-1 Prevention in African Women. In: Conference on Retroviruses and Opportunistic Infections. Seattle, USA; 2016. 110LB.
- Lanham M, Wilcher R, Montgomery ET, Pool R, Schuler S, Lenzi R, et al. Engaging male partners in women's microbicide use: evidence from clinical trials and implications for future research and microbicide introduction. J Int AIDS Soc. 2014; 17(Suppl 2):1–11. https://doi.org/10.7448/IAS.17.3.19159 PMID: 25224618
- Montgomery ET, van der Straten A, Stadler J, Hartmann M, Magazi B, Mathebula F, et al. Male Partner Influence on Women's HIV Prevention Trial Participation and Use of Pre-exposure Prophylaxis: the Importance of "Understanding". AIDS Behav. 2015; 19(5):784–93. https://doi.org/10.1007/s10461-014-0950-5 PMID: 25416076
- Doggett E, Lanham M, Wilcher R, Gafos M, Karim Q, Heise L. Optimizing HIV prevention for women: A review of evidence from microbicide studies and considerations for gender-sensitive microbicide introduction. Journal of the International AIDS Society. International AIDS Society; 2015; 18: 20536. https:// doi.org/10.7448/IAS.18.1.20536 PMID: 26700845
- Kacanek D, Bostrom A, Montgomery ET, Ramjee G, de Bruyn G, Blanchard K, et al. Intimate partner violence and condom and diaphragm nonadherence among women in an HIV prevention trial in southern Africa. J Acquir Immune Defic Syndr. 2013 Dec 1; 64(4):400–8. https://doi.org/10.1097/QAI. 0b013e3182a6b0be PMID: 24165561
- Roberts ST, Nair G, Baeten JM, Palanee-Philips T, Schwartz K, Reddy K, et al. Impact of Male Partner Involvement on Women's Adherence to the Dapivirine Vaginal Ring During a Phase III HIV Prevention Trial. AIDS Behav. 2019;(0123456789).
- Miller L, Morar NS, Kapiga S, Ramjee G, Hayes R. Prevention, Partners, and Power Imbalances: Women's views on how male partners affected their adherence to vaginal microbicide gels during HIV prevention trials in Africa. JAIDS. 2020 Dec; 85(4):458–465. <u>https://doi.org/10.1097/QAI.</u> 00000000002463 PMID: 33136745
- 40. Scorgie F, Khoza N, Baron D, et al. Disclosure of PrEP use by young women in South Africa and Tanzania: qualitative findings from a demonstration project. Cult Health Sex. 2021; 23(2):257–272. <u>https://doi.org/10.1080/13691058.2019.1703041</u> PMID: 32129720
- Mbewe L, Govender E. Male partners' influence on women's acceptance and use of PrEP products across two high HIV-burdened districts in South Africa. Afr J AIDS Res. 2020; 19(2):93–100. <u>https://doi.org/10.2989/16085906.2020.1727932</u> PMID: 32320329

- Jani N, Mathur S, Kahabuka C, Makyao N, Pilgrim N. Relationship dynamics and anticipated stigma: Key considerations for PrEP use among Tanzanian adolescent girls and young women and male partners. PLOS ONE. 2021; 16(2):e0246717. https://doi.org/10.1371/journal.pone.0246717 PMID: 33596216
- **43.** Gombe MM, Cakouros BE, Ncube G, Zwangobani N, Mareke P, Mkwamba A, et al. Key barriers and enablers associated with uptake and continuation of oral pre-exposure prophylaxis (PrEP) in the public sector in Zimbabwe: Qualitative perspectives of general population clients at high risk for HIV. PLoS One. 2020 Jan 13; 15(1):e0227632. https://doi.org/10.1371/journal.pone.0227632 PMID: 31931514
- 44. van der Straten A, Stadler J, Montgomery E, Hartmann M, Magazi B, Mathebula F, et al. Women's experiences with oral and vaginal pre-exposure prophylaxis: the VOICE-C qualitative study in Johannesburg, South Africa. PLoS One. 2014 Jan; 9(2):e89118. https://doi.org/10.1371/journal.pone. 0089118 PMID: 24586534
- Mitchell K, Wellings K, Elam G, Erens B, Fenton K, Johnson A. How can we facilitate reliable reporting in surveys of sexual behaviour? Evidence from qualitative research. Cult Health Sex. 2007; 9(5):519– 31. https://doi.org/10.1080/13691050701432561 PMID: 17687676
- Schroder KEE, Carey MP, Vanable PA. Methodological Challenges in Research on Sexual Risk Behavior: II. Accuracy of Self-Reports. Ann Behav Med. 2003; 26(2):104–23. https://doi.org/10.1207/ S15324796ABM2602\_03 PMID: 14534028
- 47. UNAIDS, AVAC. Good participatory practice Guidelines for biomedical HIV prevention trials. Geneva: UNAIDS; 2011.